Author's response to reviews

Title: Real Time Noninvasive Assessment of External Trunk Geometry during Surgical Correction of Adolescent Idiopathic Scoliosis.

Authors:

Luc Duong (duongluc@hotmail.com)
Jean-Marc Mac-Thiong (macthiong@gmail.com)
Hubert Labelle (hubert.labelle@recherche-ste-justine.qc.ca)

Version: 6 Date: 23 February 2009

Author's response to reviews: see over
Montreal, February 23, 2009

Dr Theodoros B Grivas  
Editor-in-Chief Scoliosis  
Head of Orthopaedic Department  
Orthopaedic Department "Thriasion" General Hospital  
Magula, Athens  
Greece

Re: Submission for the manuscript entitled "Real Time Noninvasive Assessment of External Trunk Geometry during Surgical Correction of Adolescent Idiopathic Scoliosis".

Dear Dr. Grivas,

The authors would like to thank the reviewers for providing relevant comments on the final version of our manuscript. Please find enclosed a revised version of this manuscript. The paper was revised according to the concerns raised prior to publication.

Thank you for accepting our paper for publication in Scoliosis, which we hope will bring new knowledge on the population affected with Adolescent Idiopathic Scoliosis.

Sincerely yours,

Luc Duong, Ph.D.  
Research Center (LIS3D)  
CHU Sainte-Justine  
3175 Côte-Sainte-Catherine Rd  
Montréal, Québec  
Canada H3T 1C5  
luc.duong@polymtl.ca

Hubert Labelle, MD  
Division of Orthopedics  
CHU Sainte-Justine  
3175 Côte-Sainte-Catherine Rd  
Montréal, Québec  
Canada H3T 1C5

hubert.labelle@recherche-ste-justine.qc.ca  
Phone: 514-345-4931 ext.4876